Author’s response to reviews

Title: Associations of the metabolic syndrome and its components with cognitive impairment in older adults

Authors:
Insa Feinkohl (Insa.Feinkohl@mdc-berlin.de)
Jürgen Janke (juergen.janke@mdc-berlin.de)
Daniel Hadzidiakos (daniel.hadzidiakos@charite.de)
Arjen Slooter (a.slooter-3@umcutrecht.nl)
Georg Winterer (georg.winterer@charite.de)
Claudia Spies (claudia.spies@charite.de)
Tobias Pischon (tobias.pischon@mdc-berlin.de)

Version: 1 Date: 28 Jan 2019

Author’s response to reviews:

Please see Response Letter uploaded as "Supplemental Material".